Author's response to reviews

Title: Antibiotic prophylaxis for caesarean section at a Ugandan Hospital: a randomised clinical trial evaluating the effect of administration time on the incidence of post-operative infections

Authors:
- Lomangisi D Dlamini (ngisi2@yahoo.com)
- Musa Sekikubo (msekikubo@yahoo.com)
- Janat Tumukunde (janattumukunde@gmail.com)
- Charles Kojjo (charleskojjo@gmail.com)
- Davidson Ocen (d4davidson@gmail.com)
- Arthur Kwizera (akwizera@chs.mak.ac.ug)

Version: 5 Date: 7 February 2015

Author's response to reviews: see over
Response to Editorial comments

Title: Antibiotic prophylaxis for caesarean section at a Ugandan hospital: a randomised clinical trial evaluating the effect of administration time on the incidence of post-operative infections

Dear Editorial Team

Thank you for assessing our paper. We hope we have addressed the comments made and changes on the manuscript are as follows:

1. The abstract has been reviewed with reference from CONSORT guidelines.
2. The trial registration and date of registration have been added at the end of abstract on page 2 lines 24 and 25.
3. The name of our International Review Board on Page 5 lines 5-7.
4. We have added a sentence to clarify that it was informed consent. Patients were given information sheet about study and it was verbally explained to them. They had to sign when they accepted to be part of study. See page 5 lines 24-26.
5. All authors have written confirmation on the noted change and emailed directly to editorial. The author whose name was later added was because when we initially submitted she had not decided as to which name and address to be put for publication yet she was part of the study from inception. We apologies for the inconvenience it has caused.
6. The CONSORT checklist has been included as additional file.

Yours sincerely

Dr Lomangisi Diana Dlamini